Author's response to reviews

Title: Motivational brief intervention for the prevention of sexually transmitted infections in travelers: a randomized controlled trial

Authors:

Nicolas Senn (nicolas.senn@gmail.com)
Serge de Vallière (serge.de-valliere@hospvd.ch)
Didier Berdoz (didier.berdoz@chuv.ch)
Blaise Genton (blaise.genton@unibas.ch)

Version: 6 Date: 31 October 2011

Author's response to reviews:

Dear Editor,

In regards to the above mentioned manuscript, we have just corrected 2 spelling mistakes and moved the name of figure 1 to the end of the manuscript.

Best Wishes
Nicolas Senn